Author's response to reviews

Title: Something Is Amiss In Denmark: A Comparison Of Preventable Hospitalisations And Readmissions For Chronic Medical Conditions In The Danish Healthcare System And Kaiser Permanente

Authors:

Michaela Schiøtz (mlsz@steno.dk)
Mary Price (Maggie.Price@kp.org)
Anne Frølich (anne.frolich@dadlnet.dk)
Jes Søgaard (jes@dsi.dk)
Jette K. Kristensen (JKK@alm.au.dk)
Allan Krasnik (a.krasnik@pubhealth.ku.dk)
Murray N. Ross (Murray.Ross@kp.org)
Finn Diderichsen (fidi@sund.ku.dk)
John Hsu (jhsu7@partners.org)

Version: 4 Date: 6 December 2011

Author's response to reviews:

To the BMC Health Services Research editors,

We have now made the final requested changes in the manuscript 1979226988542835 - Something Is Amiss In Denmark: A Comparison Of Preventable Hospitalisations And Readmissions For Chronic Medical Conditions In The Danish Healthcare System And Kaiser Permanente. Please let us know if you have any further questions or comments.

Yours Sincerely,

Michaela Schiøtz on behalf of the authors